Author’s response to reviews

Title: Sensitivity and specificity of an algorithm based on medico-administrative data to identify hospitalized patients with major bleeding presenting to an emergency department

Authors:

Emmanuel OGER (emmanuel.oger@univ-rennes1.fr)
Marie Anne BOTREL (ma.botrel@laposte.net)
Catherine JUCHAULT (catherine.juchault@chu-rennes.fr)
Jacques BOUGET (jacques.bouget@univ-rennes1.fr)

Version: 2 Date: 21 Sep 2019

Author’s response to reviews:

We modified the last version (R2) of the manuscript as follows, page 15, lines 257 to 259:

This study was authorized by the local ethics committee, named Comité d'éthique du CHU de Rennes (Rennes University Hospital), committee’s reference number 13.06. Data used in this study was anonymised before its use.

Manuscript is provided as a clean version.